Author's response to reviews

Title: Angiotensin-converting enzyme gene insertion/deletion polymorphism in migraine patients

Authors:

Erling E Tronvik (Erling.Tronvik@ntnu.no)
Lars LJ Stovner (Lars.Stovner@ntnu.no)
Gunnar G Bovim (Gunnar.Bovim@ntnu.no)
Linda LRW White (Linda.White@ntnu.no)
Amanda AJG Gladwin (Amanda.Gladwin@astrazeneca.com)
Kathryn K Owen (Kathryn.Owen@astrazeneca.no)
Harald H Schrader (Harald.Schrader@ntnu.no)

Version: 4 Date: 15 March 2008

Author's response to reviews:

1. The list of authors in the manuscript has been written exactly as they are in the submission system.
2. Each affiliation has been written in full.
3. The key words from the abstract have been removed.
4. References have been changed from superscript numbers to square brackets in the text.
5. Abbreviations have been written as a paragraph rather than as a list.
6. The tables have been edited using a Tables tool, but the content has not been changed.